
Present : Joel Closier, Davide Perego, Nick Brook, Stuart Paterson, Andrew Smith, Roberto Santinelli, 
Marcos Seco Miguelez, Angelo Carbone, Raja Nandakumar, Andrei Tsaregorotsev

Simulation
Nothing special.  All pending requests have been prepared.

Reconstruction
No job rescheduled at RAL and IN2P3. all the others sites are running.
Stagein service is tested  to solve the problem of lcg-gt problem. The job with INPUTDATA is submitted 
to the DATA optimiser the status is set to waiting for  stagein. A stagein request is send to the site and as 
soon as all the files are staged, the status of the job is changed to Waiting . 1 Thread by Tier1 site is set.

We roll back the lcg-getturls to lcg-gt and a v2r13p7 version of DIRAC is in preparation.

After the network intervention at CERN this morning, all services have been restarted to be sure that 
they are running smoothly. 

VOBox transfer
V2r13p6 is installed. Lot of queries to the top-bdii. Raja will check that on each VOBox we will use the 
local bdii and not the top one.

AOB
Nick ask to restart the reconstruction for b-incl on site which are running and to restart activity with 
stripping.

Angelo inform us that the srm-durable endpoint is set at CNAF and a news will be issue during the day.
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